Author’s response to reviews

Title: Two monogenic disorders masquerading as one: Severe congenital neutropenia with monocytosis and non-syndromic sensorineural hearing loss

Authors:
Parvathy Venugopal (parvathy.venugopal@sa.gov.au)
Lucia Gagliardi (Lucia.Gagliardi@sa.gov.au)
Cecily Forsyth (cecily.forsyth@healthemail.com.au)
Jinghua Feng (jinghua.feng@unisa.edu.au)
Kerry Phillips (Kerry.Phillips@sa.gov.au)
Milena Babic (milena.babic@sa.gov.au)
Nicola Poplawski (Nicola.Poplawski@sa.gov.au)
Hugh Rienhoff (hrienhoff@datahooks.com)
Andreas Schreiber (andreas.schreiber@adelaide.edu.au)
Christopher Hahn (chris.hahn@sa.gov.au)
Anna Brown (anna.brown@sa.gov.au)
Hamish Scott (hamish.scott@sa.gov.au)

Version: 2 Date: 22 Jan 2020

Author’s response to reviews:

Dear Editor/Assistant Editor,

We have accounted for and revised the manuscript as per the recommendations made by the assistant editor. Accordingly, we have uploaded a copy of the revised manuscript.

We have initiated the process of uploading our datasets to the European Genome-Phenome Archive (EGA). We are yet to receive accession numbers/web links and will update you as soon as we hear from EGA.

Sincerely,

Professor Hamish S. Scott
(On behalf of all authors)